Author's response to reviews

Title: Outcome and patients' satisfaction after functional treatment of acute lateral ankle injuries at emergency departments versus family doctor offices: a prospective randomized study

Authors:

Patrik R Schwab (patrik.schwab@gmx.ch)
Lorin M Benneker (lorin.benneker@insel.ch)
Stefan Eggli (stefan.eggli@insel.ch)
Heinz Zimmermann (heinz.zimmermann@insel.ch)
Aristomenis K Exadaktylos (exadaktylos@exadaktylos.ch)

Version: 6 Date: 14 December 2008

Author's response to reviews:

latest changes as required by the editorial production team (email from 8 dec 2008)